Author's response to reviews

**Title:** Echocardiographic partition values and prevalence of left ventricular hypertrophy in hypertensive Nigerians

**Authors:**

Adewole Adebiyi (wadebiyi@yahoo.com)
Okechukwu S Ogah (osogah56156@yahoo.com)
Akinyemi Aje (ajiayemi@yahoo.co.uk)
Dike B Ojji (dikeojji@yahoo.co.uk)
Adedeji K Adebayo (akbayo53972@yahoo.com)
Olulola O Oladapo (dowo@skannet.com.ng)
Ayodele O Falase (aofalase@yahoo.com)

**Version:** 4  **Date:** 27 August 2006

**Author's response to reviews:**

1. The authors list correspond to the list in the submission system
2. Title page
   - the full stop at the end of the title has been removed
   - sentence case has been used in the title
   - full stops at the end of the addresses have been deleted
   - the empty line between 1 and 2, and 6 and 7, and the first two e-mail addresses have been deleted
   - Affiliation 5 is now superscripted
   - the key word section has been removed
3. Statistical Analysis: the extra full stop at the end of the sentence finishing 'compare proportions' has been removed
4. Section headings: all section headings have been made the same text font and size.
5. Acknowledgements: Information on source of funding has been included
6. Tables - sentences case has been used for this section heading and the underlining has been removed